%12/98 Russell Brown
%1/96
\documentstyle[12pt]{article}
\pagestyle{empty}

\newcommand{\course}{MA162}
\newcommand{\sections}{???-???}
\newcommand{\daytime}{MW??:00--??:50}
\newcommand{\room}{CB???}
\newcommand{\office}{POT???}
\newcommand{\email}{???@ms.yky.edu}
\newcommand{\officehours}{???}
\newcommand{\instructorname}{??? ???}


\textwidth 6in
\oddsidemargin 0.25in
\topmargin-0.25in
\textheight 8.5in
\begin{document}

\begin{flushleft} 
MA162:\sections \hfill \instructorname\\
\daytime \hfill \office\\
\room \hfill \email\\
 Spring 1999 \hfill \officehours \\

\medskip 


Contact information for teaching assistants, if applicable

\medskip 
{\bf Instructor grade}: This may be based on quizzes, a small
amount of collected homework, attendance, class participation and a
project. Each instructor should explain how he or she will
compute the 100 points listed as instructor grade.

Policy on missed tests and assignments.

Policy on use of calculators--I suggest that calculators be
allowed. Of course, this means that you must ask the students to
present their calculations, not just the answer when they solve a
linear system.


Anything else you feel is relevant. 

\end{flushleft}
\end{document}

